Author's response to reviews

Title: Vocal local versus pharmacological treatments for pain management in tubal ligation procedures in rural Kenya: a non-inferiority study

Authors:

Sarah C Keogh (sarahckeogh@gmail.com)
Kenzo Fry (kenzo.fry@mariestopes.org)
Edwin Mbugua (edwin.mbugua@mariestopes.or.ke)
Mark Ayallo (mark.ayallo@mariestopes.or.ke)
Heidi Quinn (heidi.quinn@mariestopes.org)
George Otieno (gotiochiengs@yahoo.co.uk)
Thoai D Ngo (thoai.ngo@mariestopes.org)

Version: 2 Date: 31 May 2013

Author's response to reviews:

The manuscript has been revised to include the clinical trials registration number for the trial (registered in the Pan African Clinical Trials Registry PACTR). The registration number can be found at the end of the abstract. We look forward to your comments.